

Review



Sensors on the Internet of Things Systems for Urban Disaster **Management: A Systematic Literature Review**

Fan Zeng, Chuan Pang and Huajun Tang *D

School of Business, Macau University of Science and Technology, Taipa, Macao 999078, China; fzeng@must.edu.mo (F.Z.); cpang@must.edu.mo (C.P.) * Correspondence: hjtang@must.edu.mo

Abstract: The occurrence of disasters has the potential to impede the progress of sustainable urban development. For instance, it has the potential to result in significant human casualties and substantial economic repercussions. Sustainable cities, as outlined in the United Nations Sustainable Development Goal 12, prioritize the objective of disaster risk reduction. According to the Gesi Smarter 2030, the Internet of Things (IoT) assumes a pivotal role in the context of smart cities, particularly in domains including smart grids, smart waste management, and smart transportation. IoT has emerged as a crucial facilitator for the management of disasters, contributing to the development of cities that are both resilient and sustainable. This systematic literature analysis seeks to demonstrate the sensors utilized in IoT for the purpose of urban catastrophe management. The review encompasses both the pre-disaster and post-disaster stages, drawing from a total of 72 articles. During each stage, we presented the characteristics of sensors employed in IoT. Additionally, we engaged in a discourse regarding the various communication technologies and protocols that can be utilized for the purpose of transmitting the data obtained from sensors. Furthermore, we have demonstrated the methodology for analyzing and implementing the data within the application layer of IoT. In conclusion, this study addresses the existing research deficiencies within the literature and presents potential avenues for future exploration in the realm of IoT-enabled urban catastrophe management, drawing upon the findings of the evaluated publications.

Keywords: sensors; Internet of Things; urban disaster management; flood; earthquake; landslide; search and rescue

1. Introduction

Cities serve as the primary hubs for economic activities, social interactions, cultural expressions, and overall human existence [1]. It is anticipated that by the year 2050, approximately 86% of affluent nations will have undergone urbanization, while around 64% of developing nations will have experienced the same phenomenon [2,3]. At present, the global urban population stands at approximately 4.27 billion individuals, constituting approximately 55% of the total global population [1,4]. It is anticipated that almost 70% of the global population will undergo urbanization and relocate to urban areas by the year 2050 [4]. This significant shift will probably result in a corresponding expansion of the world's metropolitan regions, encompassing an estimated additional land area of 1.2 million square kilometers [4].

Cities often have larger population densities, making them more vulnerable to many sorts of disasters. As a result, cities have major impacts as a result of these disasters [5]. Disasters possess the capacity to cause harm to human lives and give rise to unfavorable economic and environmental outcomes [6,7]. From 2001 to 2020, there was an annual occurrence of big and medium-sized disasters ranging from 350 to 500 [8]. Furthermore, it is important to acknowledge that a greater population density results in a heightened demand for rescue services, therefore requiring more sophisticated strategies for catastrophe management and the deployment of disaster relief efforts [9]. The lack of effective



Citation: Zeng, F.; Pang, C.; Tang, H. Sensors on the Internet of Things Systems for Urban Disaster Management: A Systematic Literature Review. Sensors 2023, 23, 7475. https://doi.org/10.3390/ s23177475

Academic Editors: Ah-Lian Kor, Mohamad Hanif Bin Md Saad, Nawar Jawad and Ang Mei Choo

Received: 20 June 2023 Revised: 20 August 2023 Accepted: 23 August 2023 Published: 28 August 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

communication between public rescue and safety groups, rescue teams, first responders, and persons who are trapped worsens the situation [10]. Furthermore, it is important to acknowledge that disasters possess the capacity to inflict substantial harm against essential infrastructure systems, encompassing, but not limited to, electrical grids, water distribution networks, transportation networks, and communication systems [5]. Disasters have the capacity to disrupt economic activities and yield significant economic losses. Between the years 2008 and 2018, an extensive examination indicates that a cumulative count of 3751 occurrences of natural catastrophes took place, including a diverse range of phenomena, including earthquakes, floods, and tsunamis. The occurrence of these catastrophic catastrophes led to a significant economic downturn, resulting in a total financial loss of \$1.658 billion [9]. Urban disasters have the potential to yield substantial environmental ramifications, encompassing the release of pollutants and the handling of waste disposal. In light of the considerable repercussions that catastrophes have on urban environments,

leading to enormous losses, it is crucial to improve the management of urban disasters. The achievement of the United Nations Sustainable Development Goal (UNSDG) 12 entails the need to diminish the probability of catastrophic events and enhance the overall resilience of urban regions to withstand and recuperate from such occurrences by the year 2030 [11]. The successful attainment of the UNSDGs relies on the efficient execution of disaster management policies within metropolitan regions [12].

The concept of disaster management entails the systematic coordination and administration of various endeavors during all phases of a disaster, including but not limited to mitigation, relief, response, and recovery [9]. The primary objectives of disaster management encompass the initiation of timely alerts, the acquisition of real-time data, the precise assessment of damages, the prompt identification of evacuation pathways, and the efficient administration of emergency provisions [9]. The conventional methods of disaster management are becoming outdated due to their inability to effectively gather data from various sources in real-time and process and evaluate vast quantities of catastrophe-related information in real-time [9].

IoT enables the collection and analysis of real-time data, presenting opportunities for addressing catastrophe management in urban areas [11,13]. IoT can be described as a framework that facilitates inter-device communication via the Internet [9]. The promise of technology to facilitate complex decision support systems is evident through its ability to deliver services in a more accurate, organized, and intelligent manner [13]. IoT has significantly enhanced the capacity for analyzing catastrophe risks, namely in the areas of floods and earthquakes. This advancement has facilitated the development of more effective disaster response plans and risk management policies [11]. Numerous instances exist wherein the IoT is employed for the purpose of regular surveillance of natural occurrences, transmission of alert alerts, and provision of timely information to disaster management authorities [11]. In 2020, 23 out of 195 UN countries had effective disaster early warning systems, which successfully protected 93.63% of the population at risk from natural disasters in those countries (https://sendaimonitor.undrr.org/analytics/global-target/16/8 (accessed on 19 May 2023)). Flood warning systems can reduce flood losses by 35% annually (https://documents1.worldbank.org/curated/pt/609951468330279598/pdf/693580 ESW0P1230aster0Risk0Reduction.pdf (accessed on 19 May 2023)). Early warning systems on the West Coast of the United States enhance population preparedness for the disaster, reducing the risk of injuries by 50% [14]. The utilization of IoT technologies facilitates the acquisition of data, enabling communities to receive periodic updates and implement proactive steps in response to imminent disasters [11]. The IoT technology plays an important role in rescue actions since it can provide instantaneous updates of information [13]. The ability to make effective and precise decisions in a timely manner is crucial during relief operations due to the needs and dynamic nature of the environment [13]. The first 72 h after a disaster (i.e., the golden rescue time) are crucial for search and rescue, as the probability of finding survivors sharply decreases after this period [15]. The implementation of IoT technology has the potential to enhance the effectiveness of search and rescue

operations within a designated time frame of 72 h [15]. Hence, IoT has the capability to offer real-time monitoring, timely alerts, post-disaster response, and assistance in rescue operations, thereby assuming a significant role in urban catastrophe management. Furthermore, the utilization of IoT devices has become increasingly prevalent as a cost-effective and straightforward approach to monitoring various systems [16]. Our review aims to offer managers a comprehensive set of recommendations for the effective implementation of IoT technologies in the context of urban catastrophe management.

The architecture of IoT primarily has three layers, namely the perception layer (sometimes referred to as the sensor layer), the network layer, and the application layer [17]. Our review commences by focusing on sensors, which serve as the foundational component of the IoT framework. In the realm of IoT devices, sensors play a pivotal role in the collection and aggregation of data [18]. Sensors have the capability to be deployed in diverse environments, including riverbeds and soil. The sensors have the capability to gather and transmit data in real-time on a continuous and automated basis. Sensors are vital link between the physical and digital realms, assuming a pivotal function within the IoT framework. Subsequently, the data would be conveyed to the application layers for the purpose of data analysis and support applications, utilizing diverse communication technologies and protocols. Then, the data would be transmitted to application layers for data analysis and handle applications through various communication technologies and protocols, facilitated by gateways in the network layer [17]. The utilization of sensor-generated data inside IoT systems can facilitate data analysis and decision-making processes in the field of disaster management, provided that the data is successfully uploaded to the application layers [19]. Hence, the primary objective of this study is to address the research inquiries pertaining to both the pre-disaster and post-disaster stages.

1. What kinds of sensors are used to collect data? And what kinds of data are focused on the pre-disaster stage and post-disaster stages, respectively?

We aim to present a comprehensive analysis of the sensors employed in many catastrophe scenarios, with a particular focus on investigating the feasibility of developing universal sensors capable of addressing multiple types of disasters during the pre-disaster phase. During the stage following a disaster, it is imperative to deliberate on the appropriate data to be gathered for the purpose of post-disaster management.

2. What kinds of communication technologies and protocols are used to transmit the data from sensors?

We aim to examine the communication technologies and protocols employed during the pre-disaster stage, with the objective of identifying the prevailing and effective communication technologies and protocols. Additionally, we will examine the emergency communication technologies and protocols implemented during the post-disaster phase. The communication technologies and protocols utilized during the post-disaster phase differ from those employed in the pre-disaster phase due to the potential destruction of the communication infrastructure established prior to the occurrence of the disaster.

3. What methods were used to analyze sensor data?

The enhancement of machine learning algorithms has the potential to reduce expenses associated with sensors and facilitate expedited disaster alert systems. Furthermore, the utilization of visual algorithms facilitates the detection of various types of disasters, such as floods and earthquakes, via cameras. Therefore, it is imperative to conduct an investigation into the processes of data analysis.

4. What are the differences between the IoT technologies used in the pre-disaster and post-disaster stages?

Prior research has shown a greater emphasis on the utilization of sensors in the IoT for disaster management in the pre-disaster phase as opposed to the post-disaster phase. Esposito et al. [14] conducted a comprehensive review of early warning systems for natural catastrophes in the pre-disaster stage, specifically focusing on IoT. Ahmed et al. [20] critically examine the utilization of affordable sensors in the monitoring of climate-related disasters in coastal regions. Nonetheless, it is imperative for disaster management to encompass many activities during both the pre-disaster stage, such as disaster identification and prevention, as well as the post-disaster stage, including evacuation, search and rescue operations, and rehabilitation efforts [21,22]. Disasters can manifest abruptly or unexpectedly [6]. The likelihood of locating individuals who have survived a catastrophic event significantly decreases subsequent to the first 72 h period [15]. Efficient disaster response systems are crucial in order to mitigate human suffering and mortality rates [6]. Hence, we not only address the sensors employed in IoT systems during the pre-disaster stage. Additionally, we conduct a comparative analysis of IoT systems throughout the pre-disaster and post-disaster stages. In this study, our objective is to identify the prevailing technologies, such as sensors, communication technologies and protocols, and data analysis methodologies, that are utilized in IoT-based disaster management throughout both the pre-disaster and post-disaster stages.

The rest of the article discusses the methodology, the IoT system in the pre-disaster stage, the IoT system in the post-disaster stage, and the difference between the IoT technologies used in the pre-disaster and post-disaster stages. Section 2 introduces the methodology (i.e., systematic literature review) used in this study. Section 4 introduces the IoT system in the pre-disaster stage, while Section 5 introduces it in the post-disaster stage, which answers research questions 1–3. Section 6 compares the differences between the IoT technologies used in the pre-disaster and post-disaster stages, which answers research questions 4. Section 7 concludes the findings of this study, indicates the limitations, and shows the future research directions in this research area.

2. Methodology

Systematic Literature Review (SLR) is a way to synthesize research findings in a systematic, transparent, and repeatable manner for identifying and critically evaluating relevant research to answer specific research questions or hypotheses [23]. Previous studies widely used the SLR approach to organize literature and perform a thorough literature review in the areas of sensors in the IoT [19,20]. Regarding the SLR process, we should first collect the target literature, then identify the literature that meets the pre-specified inclusion criteria, and finally provide solid findings.

To search the target publications, we combined the keywords into three parts: (1) the IoT keywords, (2) the keywords related to "urban", and (3) the keywords related to natural disasters. Firstly, we target publications related to IoT through the keywords "Internet of Things", "wireless sensor networks", and "Internet of Everything" and their abbreviations [14,24]. Further, we scoped the publications in the scenarios of the city through the keywords. "urban", "city", and "cities" [25–28]. Finally, we scoped the publications related to disaster management through the keywords "disaster", "natural hazard", "flood", "landslide", "earthquake", "storm", "hurricane", "wildfire", "tornadoes", "cyclones", "drought", "tsunami", "typhoon", "avalanche", "heatwave", "volcan*", and "gully erosion" [8,29–36]. Therefore, our final search string was "TS = ("internet of thing*" OR "IoT" OR "IoTs" OR "wireless sensor network*" OR "WSN" OR "WSNs" OR "Internet of Everything" OR "IoE") AND TS = (urban OR city OR cities) AND TS = (disaster* OR "natural hazard*" OR flood* OR landslide* OR earthquake* OR storm* OR hurricane* OR wildfire* OR tornado* OR cyclone* OR drought OR tsunami* OR typhoon* OR avalanche* OR heatwave* OR volcan* OR "gully erosion*")". We used quotation marks to search the whole specific phrase, such as "Internet of Things". But the quotation marks prevented WoS from searching US and UK spelling variations automatically (http://webofscience.help.clarivate.com.libezproxy. must.edu.mo/en-us/Content/spelling-variations.html (accessed on 19 May 2023)). Thus, quotation marks are not necessary for a single word, such as "urban". We added an asterisk to prevent variations of words from being missed (http://webofscience.help.clarivate. com.libezproxy.must.edu.mo/en-us/Content/search-operators.html#Search (accessed on

19 May 2023)). It is worth noting that the asterisk is not available to search US and UK spelling variations (http://webofscience.help.clarivate.com.libezproxy.must.edu.mo/en-us/Content/spelling-variations.html (accessed on 19 May 2023)).

We ran the search string on the Web of Science, which is one of the significant bibliographic databases [35]. We selected Web of Science since it allowed us to select a large number of highly credible publications with impact factors [35]. We only considered English publications [35]. The types of publications include research articles, conference papers, and review papers [8]. Finally, we gathered 502 publications.

We excluded irrelevant publications through two stages: screening the title and abstract of the publications and a full review of the publications. After screening the titles and abstracts of the publications, we excluded 400 irrelevant publications. After fully reviewing the publications, we excluded 30 irrelevant publications. Finally, we constructed a database with 72 publications for a systematic literature review. Among 72 publications, 47 focus on the pre-disaster stage, while 20 focus on the post-disaster stage. Five publications both mention IoT systems used in the pre-disaster and post-disaster stages.

3. Descriptive Analysis

Figure 1 shows the distribution of sample publications by year. Academic attention was directed towards this particular research domain in the year 2007. However, there was a limited number of papers that concentrated on this specific subject field throughout the subsequent decade. The quantity of publications reached its highest point in 2021, with a total of 17 publications, but experienced a significant decline to 7 publications annually in 2022. The dataset of articles for the year 2023 is still incomplete, as it only includes items downloaded up until May 2023.





Table 1 shows the top six distributions of publications by sources. Among 72 publications, most were published in the journal *Sensors* (five publications). Two publications were published at the 2019 5th IEEE International Smart Cities Conference, Applied Sciences, Earth Science Informatics, IEEE Access, and Materials Today-Proceedings, respectively.

| Source | Number of Publications |
|--|---------------------------|
| Sensors | 5 |
| 2019 5th IEEE International Smart Cities Conference (IEEE ISC2 2019) | 2 |
| Applied Sciences | 2 |
| Earth Science Informatics | 2 |
| IEEE Access | 2 |
| Materials Today-Proceedings | 2 |

Table 1. Top six distribution of publication by sources.

4. IoT Systems in the Pre-Disaster Stage

IoTs have various advantages, such as low cost, low energy consumption, access to harsh environments, and simple installation [37]. More importantly, the sensors used in IoT systems can adapt to changes in the environment and collect real-time, high-precision environmental data [37]. Therefore, IoT systems are useful tools for monitoring natural environments and disaster management [37].

4.1. Sensors

Sensors used in the pre-disaster stage mainly collect environmental data. Accelerometers are usually used to detect earthquakes [38,39]. Except for accelerometers, more sensors are used to detect landslides, such as inertial sensors, bar extensometers, and borehole inclinometers. More publications focus on flood monitoring. Floods may occur more frequently in cities. One reason is that urban drainage systems often become saturated due to prolonged and intense rainfall [40]. Regarding flood monitoring systems, scholars use more sensors, such as rain gauges [40,41], water level sensors [40,42–47], water pressure sensors [41,45,47], cameras [41,48,49], soil moisture sensors [42], weather sensors [42], drones with drones [42], water presence sensors [44,50], temperature sensors [50], and a triaxial accelerometer [50]. The sensors are usually powered by solar batteries [41,50]. To save the energy cost of the sensors, Biabani et al. [51] introduced a model with a harmony search algorithm and improved hybrid Particle Swarm Optimization to select cluster heads. Based on Particle Swarm Optimization, they developed a multi-hop routing system with enhanced tree encoding and a modified data packet format. The single computer boards are the Raspberry Pi [43,48] and Arduino [43,44]. We will further discuss the sensors used for different disaster types in the following content.

4.1.1. Earthquake

We commonly use accelerometers to detect earthquakes. Accelerometers include triaxial accelerometers and dual-axis accelerometers. Regarding triaxial accelerometers, we can choose the accelerometer ADXL362 [39] for low-power use, while we can choose the accelerometer EPSON M-A351AU [39] and the accelerometer LSM9DSO [16] for high-precision use. Regarding dual-axis accelerometers, we can choose the accelerometer ADXL203, which is low-power and high-precision [52]. Also, we can choose triaxial accelerometers, such as L1S3DSH sensors (manufactured by STMicroelectronics) and EpiSensors [53]. The L1S3DSH sensor is ultra-low-power and high-performance [53]. The accelerometers are placed on the object being detected (e.g., buildings or bridges). The single computer boards include Raspberry [38], CC2420 DBK [52], and Sparrow v4 [16]. The microprocessors include the ATmega128L [52], ATmega128RFA1 [16], and ARM processor [53]. Sensors should be equipped with antennas to enable data transmission over long distances [39]. The sensors should be energy-saving [39]. Regarding batteries, we can choose d-cell batteries [39] and CR2032-3V lithium-ion batteries [16]. The sensors should sleep when they do not need to collect data [39]. In addition, the sensors should be low-cost [39]. For example, each sensor in Siringoringo et al.'s [39] earthquake detection system costs 2300 USD.

In addition, some scholars may use other sensors for earthquake detection. For example, Castelli et al. [54] also combined triaxial velocimeters and ultrasonic measurements to build the earthquake early warning system. Tudose et al. [16] combined an LSM9DSO-16 bit high-resolution triaxial accelerometer with an SI7020 humidity and temperature sensor, a three-axis gyroscope, a triaxial magnetometer with embedded FIFO, an SI1145- infrared proximity detection, a high-precision altimeter, a UV and ambient light sensor, and a MPL3115A2-pressure and temperature sensor.

4.1.2. Landslides

People detect landslides with more sensors, such as inertial sensors [55], accelerometers [55], bar extensometers [56,57], borehole inclinometers [56], rainfall sensors (e.g., rain gauge [58]), and displacement meters [58]. The models of inertial sensors include the IMU6050 [55]. The models of accelerometers include LIS3331LDH [55]. The models of microprocessors include the ESP32 [55,59]. The models of single computer boards include the Waspmote PRO board [55]. Usually, the sensors can store data locally on SD cards [55]. Batteries [55–57] and solar [56] are the major power sources. To improve energy efficiency, Zhang [60] designed a wavelet-based sampling process for landslide sensors. This process allowed the sensors to reduce data gathering while maintaining performance and system reliability, which allows the battery to run continuously for 3–5 months without recharging during the monsoon period. Wang et al. [57] suggested using the WorkStop recycling control mode in the batteries.

4.1.3. Floods

People develop flood detection systems based on more considerations such as rainfall [40–42,61–64], water level [40,43,44,46,47,62,63,65–68], water pressure [41,44,45,47,50,69], soil moisture [42], solar radiation [42], vapor pressure [42], relative humidity [42], humidity [42], temperature [42,61,62,70], air pressure [42,62], wind speed [42,61], wind gust [42], wind direction [42], tilt [42], lighting [42], lighting average distance [42], the flow velocity [42,62]. Moreover, Ragnoli et al. [50] also used GSM to detect locations in their flood monitoring systems.

We usually collect rainfall data using rain gauges [40,41], such as double-tipping buckets [40]. We usually measure water level with water level sensors, such as radar level sensors [40,65], ultrasonic sensors [42,47,65,68,70], and force-sensitive resistors [47]. Ultrasonic sensors include MaxBotix MB7066 [70], HC-SR04 [68], and so on. A method to measure water level is to measure water pressure and convert that data into water level [45,64,69]. Ragnoli et al. [50] detected water with electrical resistance. To be specific, they used fork-shaped probes made of conductive and corrosion-resistant metal with 10 cm long and 1.5 cm spaced terminals [50]. The resistance would drop when the terminal came into contact with water [50]. Regarding fault tolerance, they added a triaxial accelerometer (i.e., ADXL345) to use when the water sensors were damaged [50]. Malik et al. [44] combined the Adafruit SHT31-D temperature and humidity sensor with dual ultrasonic sensors to monitor water levels. Another sensor type aims to detect the presence of water. Malik et al. [44] combined a waterproof temperature sensor with dual water presence sensors to detect water presence. Mendoza-Cano, Aquino-Santos, Lopez-de la Cruz, Edwards, Khouakhi, Pattison, Rangel-Licea, Castellanos, Martinez-Preciado, Rincon-Avalos, Lepper, Gutierrez-Gomez, Uribe-Ramos, Ibarreche and Perez [42] combined Drifters and river drones to measure parameters such as river flow velocity and water temperature during the flooding events. Drifters were the main measuring tool, while RiverDrone aimed to locate the Drifters. Mousa, Oudat, Claudel and Ieee [70] suggested measuring temperature through passive infrared sensors (e.g., Melexis MLX90 614).

Cameras are also effective tools for flood detection [41,48,49,63,69,71–73]. Regarding the use of cameras, Castro et al. [72] suggested using no infrared cameras, while Castro et al. [72] and Garcia et al. [48] suggested using cameras with water level markers. Regarding water level markers, Castro et al. [72] suggested placing highly visible reflective tapes on surfaces visible to cameras ranging from 0 to 1.5 m. Each tape was spaced out with other tapes, which allowed us to obtain a better approximation of the severity of the water level. This method could improve accuracy because it was not affected by the temperature and humidity of the air or the objects that could absorb wave sounds. In addition, Garcia et al. [48] suggested putting visible marks on the image captured by the camera to detect the flood-severity level.

The types of power supplies are numerous. For example, we can place sensors on electricity poles to absorb power [40,47]. Solar and/or batteries are the common power supply [41,44,50,70], such as Seeed Studio solar cell batteries with TP4056 charge regulators [50] and solar-powered Lithium Iron Phosphate batteries [70]. The models of microprocessors include the Raspberry Pi 4 Model B [48], Arduino Uno [43], Raspberry Pi 3 Model B+ [43], Analog to Digital Converter [50], Intel PXA271 XScale [71], and ESP32 [67]. The models of single computer boards include Arduino [44,68], NodeMCU [47], TelosB [74], Raspberry Pi [72], ARM Cortex M4 [70], and Arduino DUE [67]. We can also consider SD cards for local data storage [70].

4.1.4. Others

Park and Baek [75] introduced the detection of heatwaves and cold waves. Alhamidi et al. [76] presented an IoT-based tsunami monitoring system. Aljohani and Alenazi [77] introduced a storm detection system. We can use meteorological sensors to detect heatwaves and cold waves by monitoring parameters such as temperature, relative humidity, noise, illumination, ultraviolet, vibration, PM10, PM2.5, wind speed, wind direction, CO, NO₂, SO₂, NH₃, H₂S, and O₃ [75]. Alhamidi, Pakpahan, Simanjuntak and Iop [76] used the ADXL335 accelerometer to read vibrations in the seafloor crust because tsunamis are caused by vibrations and faults in the seafloor crust. They also connected sensors to flare-marking buoys to provide information to the nearest disaster mitigation center. They used the Arduino Uno as a single computer board. Regarding storm detection systems, Aljohani and Alenazi [77] suggested using weather sensorsincluding humidity and lightningsensors. For forest fires detection, Viegas [78] used sensors to collect data such as temperature, humidity, gas concentrations, rain rate, wind direction, and wind speed. They also use cameras, including Pan-Tilt-Zoom and Fixed Cameras, and Unmanned Air Vehicles equipped with cameras. To detect typhoons, Wang et al. [79] suggested using meteorological satellites as sensors to obtain high-resolution remote sensing image data to recognize typhoon clouds and locate the typhoon center.

4.2. Communication Technologies and Protocols

People usually transmit the data from sensors to servers through Bluetooth [38], Ethernet [38,42,43], Wi-Fi [38,43,55,56,59,68,72], and cellular communication technology [42,57,59,71,80], Radio Frequency [42], and radio [44]. Cellular communication technologies include GSM [57], GPRS [57,71], and 3G [42]. The communication protocols include Choco protocol [39], UDP [41], IPv6 with LoWPAN [41], IEEE 802.15.4 [16,41,52], Message Queuing Telemetry Transport (MQTT) [42,43,47,59,67,81], concurrent multipath transfer protocol [49,82], LoRaWAN [50,62,80], TCP/IP Internet protocol [50,83], Hyper Text Transfer Protocol (HTTP) [61], Zigbee [16,45,71], LRWiFi [59], Cat-M1 [59], CoAP [59], XBee [70]. The data is usually transmitted in JSON format [42,50,61]. Regarding data storage, people may use local data storage (e.g., SD memory cards [55]) and cloud storage (e.g., MongoDB [48], Dynamo [43]). Miao and Yuan [58] used the SQL Server 2008 database software. Malik et al. [44] store the data in Oracle's MySQL and host the database on an Ubuntu Server. Drones can relay data from sensors to base stations, thereby effectively achieving large-scale data transmission [18]. Drones can cooperate with drifters to collect river velocity data. In this combination, drones aim to locate the drifters and transmit the data from the drifters to a server [42].

The connection solutions for earthquakes include Bluetooth [38], Ethernet [38], Wi-Fi [38], Choco protocol [39], IEEE 802.15.4 [16,52], MQTT [81], and Zigbee [16]. Regarding the connectivity solutions for landslides, we can use Wi-Fi [55,56], cellular communication technology [57,59], MQTT [59], LRWiFi [59], Cat-M1 [59], and CoAP [59]. The connection solutions for floods include UDP [41], IPv6 with LoW-PAN [41], IEEE 802.15.4 [41], MQTT [42,43], Ethernet [42,43], cellular communication technology [42,71,80], Wi-Fi [43,68,72], concurrent multi-path transfer protocol [49], LoRaWAN [50,62,80], TCP/IP [50,83], radio [44], HTTP [61], ZigBee [45,71], 6LoW-PAN [74], and XBee [42,70]. Concurrent transfer can achieve higher throughput [49,82], accelerate transmission [49,82], reduce packet loss [49,82], and save energy [39]. Choco protocol [39] and 6LoWPAN [74] can save energy. In addition, Luo et al. [84] proposed the "MWAC model" for sensor networks to save power and transmit information over long distances (p. 49).

Usually, we convert information between different protocols using different systems and intermediate devices. For instance, Ragnoli et al. [50] and Gomes et al. [83] both transmitted the data for sensors to the server via TCP/IP while transmitting the data from the server to user applications through HTTP. However, Ferraz et al. [61] built the servers to connect sensors and human clients based on the HTTP protocol without the use of other protocols.

4.3. Analysis and Applications of Sensor Data

Firstly, we tend to emphasize that data pre-processing is important to improve the efficiency of data analysis. Some missing sensor data were recorded as zero because of the irregular data transmission and the irregular observation time. We could not distinguish this missing data from an observed zero value. To solve this problem, Park and Baek [75] suggested some quality management for sensor networks, such as data pre-processing (time allocation and filling short gaps in missing data), physical limit check, climate range check, internal consistency check, persistence check, step check, spatial consistency check, spatial outlier check, and data reconstruction using spatial and temporal gap-filling. In addition, Wang and Abdelrahman [62] suggested using a divide-and-conquer approach to process the high-dimensional data inputs from sensors. For example, we can group the sensors by their physical locations and customize the model to process each sensor.

Since sensors have limited resources, another way to improve the efficiency of data analysis is to combine fog computing and cloud computing [41,85]. To be specific, sensors send the data to the fog periodically [85]. After the fog pre-processes the data, it will be transmitted to the cloud [85]. Therefore, fog computing is mainly responsible for concentrating, distributing, caching, and analyzing the data, detecting abnormalities, analyzing the data on a smaller scale, sending notifications and feedback, and forwarding summarized data to the cloud periodically [41,85]. Fog computing can reduce the latency of the service, respond to any emergency change immediately, and reduce the burden on the cloud [41,85]. Cloud computing is responsible for combining and permanently storing all the data in the system to obtain a general view of the monitored environment [41,85]. In addition, the cloud accumulates some historical data over time, which can provide important information about the weather in each area [85]. We can also run machine learning algorithms on the historical data to form a smart classifier [85]. In a word, we can use fog computing to analyze a small range of data for the timely detection of disasters and warnings. Cloud computing builds long-term predictive models by analyzing data on a larger scale. In addition, cooperation between fog computing and cloud computing can improve the efficiency of data analysis through data pre-processing in fog computing and database construction in cloud computing.

Regarding data analysis methods, we can use some advanced techniques, such as machine learning [38,45,62,70,85–87], deep learning [46,73], and time-series data analysis [58]. Some studies may use traditional methods, such as mathematical modeling (e.g., Markov Process, Laplace Transformation) [68], observing the signal (e.g., flare marker buoys) [76], and comparing the current situation with past cases [84]. Table 2 shows data analysis methods for different types of disasters.

 Table 2. Data analysis method used in analyzing data from sensors.

| Types of Disasters | Method | Detail | Reference |
|--------------------|---|--|---------------------------|
| Earthquake | Machine learning | Convolutional Neural Network Recurrent Neural Network | Kim et al. [38] |
| Landslide | Time-series data analysis | Grey System Forecasting | Miao and Yuan [58] |
| Floods | Machine learning | Bayesian Learning Multi-Layer Perceptron Artificial Neural Networks Random Forest J-48 Decision Tree Random Tree Simple Cart Decision Tree BFTree | Furquim et al. [45] |
| Floods | Machine learning | Artificial Neural Networks LSTM | Wang and Abdelrahman [62] |
| Floods | Machine learning | Random Forest Decision Tree KNN | Aljohani et al. [85] |
| Floods | Machine learning | Artificial Neural Network | Mousa et al. [70] |
| Floods | Machine learning | Artificial Neural Network | Goyal et al. [87] |
| Floods | Deep learning | Deep Neural Network | Junior et al. [73] |
| Floods | Time-series data analysis + Machine learning | BiGRU Neural Network + Attention Mechanism | Chen et al. [86] |
| Floods | Time-series data analysis + Machine learning | Multilayer Perceptron artificial neural network | Furquim et al. [37] |
| Floods | Time-series data analysis + Machine learning | Multilayer Perceptron artificial neural network | Furquim et al. [64] |
| Floods | Time-series data analysis + machine learning | Multilayer Perceptron artificial neural network | Furquim et al. [69] |
| Floods | Image processing algorithm | | Garcia et al. [48] |
| Floods | Image processing algorithms | Edge Keeping Index SURF. | Liu et al. [71] |
| Floods | Image processing algorithm | Color segmentation Morphological operations Shape detection | Castro et al. [72] |
| Floods | Image processing algorithm | DNN Pruning Algorithm + Randomized Heuristic | Junior et al. [73] |
| Floods | Mathematical modeling | Markov Process Laplace Transformation | Tyagi et al. [68] |
| Floods | Data retrieval | Compare current situations with past cases | Luo et al. [84] |
| Typhoon | Image processing algorithm | Attention Mechanism Fast R-CNN Transfer Learning method | Wang et al. [79] |
| Tsunami | Observation | Flare marker buoys | Alhamidi et al. [76] |

Machine learning techniques are commonly used to analyze the data from sensors. We can use machine learning techniques to analyze the data related to different types of disasters, such as earthquakes [38] and floods [45,62,70,85–87]. The machine learning

techniques used in earthquake detection include convolutional neural networks [38] and recurrent neural networks [38]. They can analyze the data collected by accelerometers. The machine learning techniques used in flood detection include Bayesian Learning [45], Multi-Layer Perceptron Artificial Neural Networks [45], Random Forest [45], J 48 Decision Tree [45], Random Tree [45], Simple Cart Decision Tree [45], and BFTree [45]. They can classify and analyze the water level data. Regarding data classification, we can classify the water level data into stable level (i.e., -20° , 20°), slight increase level (i.e., 20° , 45°), high increase level (i.e., 45° , 90°), slight decrease level (i.e., $(-20^{\circ}, -45^{\circ})$ and high decrease level (i.e., -45° , -90°) [45]. Another classification is 0 to 0.25 for Mild level, 0.26 to 0.5 for the Moderate level, 0.51 to 0.75 for Severe level, and 0.76 to <1 for Critical level [87].

Furquim et al. [37] assumed that the water level data is time-series data. They modelled the time series data based on chaos theory. They used the false nearest neighbor method to estimate the value of the separation dimension and the embedding dimension. Furquim et al. [64] found that the best results for all the sensors were obtained when the separation dimension was one and the embedding dimension was two. When we adopted the distributed approach, MLP could present the peak values in a better way. The peak values are very important in examining flood prediction since they are at the points where the flooding occurs. In addition, Furquim et al. [69] used a multilayer perceptron artificial neural network to construct the recursive prediction model and obtained better results when the separation dimension was one and the embedding dimension was four. Furthermore, Chen et al. [86] suggested using a Bidirectional Gated Recurrent Unit (BiGRU) model with attention mechanisms to deal with the time-series flood data. The attention mechanism is used to automatically adjust how well the input features match the output features, while the BiGRU model aims to process the input series from both directions of the time series (chronologically and anti-chronologically) and then merge their representations together.

Since we use cameras for disaster detection more commonly, image processing algorithms have become one of the most common data analysis methods [48,71–73,79]. Previous studies adopted image-processing algorithms to detect floods [48,71–73] and typhoons [79]. Although some of them are machine learning algorithms, the data they processed was different from the machine learning mentioned above. Image processing algorithms focus on image data, while the machine learning algorithms mentioned above focus on numerical data.

After data analysis, scholars tend to present the results to the public through web applications [40,42–44,47,48,57,58,61,63,65,71,72,83,85,88,89]. Some of them are mobile applications such as Ferraz et al. [61]. In addition, we usually visualize data with maps to present environmental data and disaster positions in web applications [40,42,45,48,58,63,72,78,81,83,84,88]. Interestingly, Kanak et al. [81] integrated data from the sensors to create a virtual reality environment to help residents perform fire and earthquake escape drills. In addition, email [43,50,71], messages [43,71], and social media [40,68] are the ways to send out warnings to people in time. They are also the ways to send out disaster warnings.

5. IoT Systems in the Post-Disaster Stage

5.1. Sensors

Excepted for environmental data [15,90–93], the sensors in the post-disaster stage mainly collect human health data [90,92–94] and position data [15,38,93–95], which can improve the efficiency of search and rescue.

Regarding environmental data, Ochoa and Santos [15] suggested using sensors to collect environmental data in terms of weather, chemicals, and movement. Sahil and Sood [90] placed sensors on the buildings and in-pavements in the disaster-affected areas to collect environmental data, including water level, tilt in structures, temperature of buildings and ambient, smoke detection, obstacles in the path, visibility range, and location of the sensor. Korkalainen et al. [93] suggested using gas sensors to monitor air quality. Usually, multiple agencies participate in rescue operations. Each agency

could use the sensors deployed in cities (e.g., weather stations, traffic cameras, wind sensors, precipitation meters, road surface condition sensors, and visibility meters) to collect environmental data [91,92]. Also, the agency integrated its own sensors, such as unmanned vehicles equipped with GPS sensors, acoustic detection, distance measurement, and motion sensing, and drones equipped with GPS sensors and cameras [91,92]. Regarding disaster mitigation, Goyal et al. [96] and Rahman et al. [47] collected water level data to open the floodgates. Rahman et al. [47] used ultrasonic sensors and a force-sensitive resistor to measure the water level. In addition, floods may cause potholes on the road. Ulil et al. [80] developed pothole monitoring systems with a modem accelerometer and gyroscope. They used the Raspberry Pi as a single computer board. They placed the sensors on the vehicles. For building structure health detection after an earthquake, Antonacci et al. [97] suggested using the LIS344ALH accelerometer, TAOS 2561 light sensor, and SHT11 temperature and humidity sensors, which were installed on the Imote2 platform and ISM400 board.

The human health data, including the rescuers' health data and the stranded people's health data, Boukerche et al. [94] suggested that command posts should guarantee the safety of first responders through the body-worn sensors in wearable smart devices, such as smart glasses and smart watches. Sahil and Sood [90] developed an IoT system to prioritize the evacuation of panicked, stranded people and provide them with timely medical support. They used the health sensors in stranded people's personal mobile communication devices to collect health data, including heart rate, breath rate, dizziness, sweating, chest pain, trembling, chills, choking, nausea, and the location of the individuals. In the rescue operations suggested by Johnsen et al. [92], each rescuer was equipped with a personal sensor system (e.g., a tactical vest or tactical underwear) to monitor their health and position. The sensors in the tactical vest aimed to monitor water levels. The tactical underwear contained medical sensors, including one muscle activity sensor, pulse oximeters, and a heart rate sensor, which were installed on the Arduino single computer board. Regarding detecting epidemics in disasters, Ehsani et al. [98] used the case of detecting COVID-19 in earthquakes to explain an IoT framework. In evacuation centers and temporary hospitals, they used thermal and infrared sensors to monitor people's body temperatures and detect fever. And they used heart rate sensors to measure oxygen levels and detect breathlessness. In affected areas, they monitored body temperature data through medical infrared thermometer guns. And they also deployed thermal sensors on unmanned aerial vehicles to monitor people's body temperature. Korkalainen et al. [93] used CO sensors, CO₂ sensors, optical sensors (e.g., LWIR cameras, visual range cameras), vibration sensors, sound sensors, and an ion mobility spectrometer to detect and locate life. An ion mobility spectrometer could detect volatile organic compounds, such as ammonia and Acetone [93].

The position data includes the stranded people's position data, the rescuers' position data, and the rescue vehicles' position data. Firstly, people's personal devices are effective sensors to help us collect stranded people's position data [38]. We can ask stranded people to wear radio frequency identification bracelets to locate them as well [99]. And we can locate evacuated social vehicles through portable on-board radio frequency identification tags [99]. Suri et al. [91] also suggested using traffic cameras to search for and locate people trapped in vehicles or rubble. The traffic camera can clearly take images for every passing car, and recognize the license plate and driver characteristics automatically [99]. Furthermore, we also installed the sensors in the rescuers' personal devices (e.g., tactical vests, tactical underwear) to locate the rescuers [92]. Ochoa and Santos [15] suggested using GPS, Radio Frequency positioning, and inertial sensors to track rescuers in the field. For rescue vehicles (e.g., ambulances, fire engines, police cars, and engineering vehicles), we should equip them with IoT equipment to locate them. For example, Anagnostopoulos et al. [95] use sensors to collect the position data of the emergency medical service system. Last but not least, Rahman et al. [47] installed GSM to locate the floodgates.

5.2. Communication Technologies and Protocols

Normal communication technologies and protocols are also suitable for communication in the post-disaster stage, such as Wi-Fi [38,80,93,98], Bluetooth [98], Internet [98], MQTT [47,80,92,100], LoRa/LoraWAN [92], WPAN [93], 3G/4G [80], and COAP [100]. However, disasters may destroy the infrastructure in the cities [94]. We may address this issue by maintaining the efficiency of existing communication infrastructure and using additional mobile communication tools.

To maintain communication efficiency in the face of a reduced number of communication facilities, we should develop resilient communication networks and reduce contention during data communication. Alvarez et al. [101] suggested using the Bluetooth Mesh emergency network to utilize the remaining sensors to mediate device-to-device communication in the post-disaster stage. Regarding energy-constrained IoT sensors, Ai-Turjman [102] suggested using the Cognitive Energy-Efficient Algorithm (CEEA). The CEEA was a topology-independent protocol that can handle randomness in IoT networks. The CEEA determined the path from routing nodes to sensors based on the remaining energy of each node. To be specific, the CEEA would control the remaining energy of neighbors of recent routing nodes each time before sending data from the recent routing nodes. If the energy of one of the neighboring routing nodes was less than half of its initial value, the CEEA might determine a new path to transmit the data. If the residual energy of all neighboring routing nodes are found to be below 50% of the beginning energy, the CEEA uses the same strategy. However, it was noted that multitier IoT networks and cluster- or tier-wide synchronization were the two assumptions for the effective use of the CEEA. Aljohani and Alenazi [77] introduced a multi-path resilient routing system based on software-defined networking (SDN), which combined aided-multipath routing with the capabilities of SDN. Campioni, Lenzi, Poltronieri, Pradhan, Tortonesi, Stefanelli, Suri and Ieee [100] developed a multi-domain Asynchronous Gateway of Things to enable discovery across different communication protocols and administrative domains in post-disaster relief. In addition, epidemic protocols can solve the contention caused by the reduced numbers of communication infrastructure in the post-disaster stage. Tan et al. [103] proposed an adaptive probabilistic epidemic protocol that can effectively suppresses redundant messages and reduces contention/collision levels. This protocol allowed a node to decide whether to respond to a broadcast based on information such as the number of neighbors of the broadcasting node. Ochoa and Santos [15] suggested using epidemic routing algorithms and spray and wait routing algorithms to support the dissemination of shared information among personal devices.

In addition, we can use additional mobile communication tools in post-disaster relief, such as drones [90,94,103] and vehicles participating in relief [10,22]. Tei et al. [22] proposed an opportunistic data dissemination protocol. They proposed facilitating the transmission of information from victims to rescue agencies by using the existing vehicles, including ambulances, police cars, and fire trucks, as well as the sensors inside the network. We can store the data from drones in the cloud [90]. The vehicles participating in relief include dynamic vehicles (e.g., fire trucks, ambulances) and stationary vehicles [15]. Dynamic vehicles aim to transmit data from sensors to stationary vehicles, while stationary vehicles (e.g., the base vehicles) focus on the transmission between different teams or companies in post-disaster relief [15]. Further, Johnsen, Zielinski, Wrona, Suri, Fuchs, Pradhan, Furtak, Vasilache, Pellegrini, Dyk, Marks, Krzyszton and Ieee [92] suggested using the base vehicle as a central server that utilizes Wi-Fi to receive data from sensors deployed on mobile unmanned vehicles and in cities. The video data from the drones could also be transmitted to the base vehicle through radio links [92].

5.3. Analysis and Applications of Sensor Data

People focus on analyzing position data and health data in order to plan evacuation routes [38,90], allocate ambulance vehicles [95], understand the health of trapped individuals [90], and integrate information for rescuers [91,92]. For example, Ehsani et al. [98]

can detect COVID-19 cases in a disaster by analyzing people's temperatures with machine learning techniques. Regarding locating people in disasters, Kristalina et al. [104] used the least squares method to improve the generalized geometric triangulation scheme, which allows sensors to track the position of rescuers or victims. Similarly, Konomi et al. [105] proposed a cooperative location inference mechanism to locate the sensors automatically. And they developed a user-participatory sensing environment that allows people to collect position data from sensors. In addition, we may carry out post-disaster activities with the cooperation of multiple agencies [9]. Thus, we need to collect data from various heterogeneous sensors. Konomi, Wakasa, Ito and Sezaki [105] proposed a novel multi-factor cost model to integrate the multi-modal sensor data consistently and flexibly. Li et al. [106] introduced a semi-automated role mapping process for dynamic cross-domain accesses of sensors in post-disaster relief to solve heterogeneity and protect sensitive information.

In addition, the combination of fog computing and cloud computing can improve the efficiency of data analysis [90–92]. For example, Suri et al. [91] proposed the Sieve, Process, and Forward (SPF) Fog-as-a-Service platform to address the scenario of post-disaster relief. Fog computing is also helpful in reducing the time of search and rescue since it can improve the efficiency of data analysis through pre-processing some data analysis, such as data categorization and data novelty analysis [90,91]. Cloud computing aims to store the data and process deeper data analysis [90–92,98]. The fog layer exists in the gateway (e.g., drones and evacuation vehicles) and serves as a bridge between the sensor layer and cloud layer of the Internet of Things [90,91]. The utilization of the fog layer is attributed to its position awareness and close proximity to the sensors [90]. This enables it to perform essential data pre-processing tasks, such as data categorization, novelty analysis, panic health status classification, and alarm creation [90]. Due to the inherent limitations in computing and storage capacities of the fog layer, the cloud layer was employed to store and analyze environmental and health data, as well as the corresponding panic health status data [90]. This facilitated the generation of alerts in the form of compiled medical records [90]. The Cloud layer includes temporal data mining, cloud storage, panic health sensitivity monitoring, evacuation strategy, and evacuation map building [90]. Thus, we usually develop an application to combine the data and conduct big data analysis on the cloud [91,92]. And we will place the mobile decision-making centers for post-disaster rescue on the cloud [91,92].

Regarding data application, scholars usually publish the results of data analysis on the web application [15,38,88,91,92,99,105]. Some of them are mobile applications [15,92,105]. We visualize data with maps to present the collapse, the traffic, the shelters, and the evacuation route in the post-disaster stage [15,38,88,90–92]. For example, Lwin et al. [88]'s City Geospatial Dashboard can provide road congestion information to help disaster response teams estimate travel times to reach disaster areas. toRoute planning is one of the major applications in the post-disaster stage [38]. Based on position data, Kim et al. [38]'s system can provide a real-time evacuation route guidance service, including searching for the safest shelter and showing pedestrian paths for users. Then, previous studies tend to use dynamic programming methods to support route planning. Konomi et al. [105] developed a user sensing environment to collect geo-tagged sensor data, omnidirectional cameras, and environmental sensors (e.g., temperature, humidity) to solve dynamic routing planning problem in the post-disaster stage. Liu and Wang [99] suggested using the variable structure discrete dynamic Bayesian network model for real-time dynamic path planning This approach could facilitate the prompt evacuation of social vehicles from the event area, while also ensuring the timely arrival of rescue vehicles at their allocated position to carry out necessary tasks. Anagnostopoulos et al. [95] developed a real-time dynamic routing algorithm for the ambulance arrangement to reduce the time per route, distance covered, and fuel consumption.

Some scholars designed mitigation measures specifically for floods. Goyal et al. [96] use reinforcement learning to develop gate-control systems on Flash Flood Bypass Waterways to evacuate the flood water to channels. Rahman et al. [47] also controlled the

15 of 21

valves based on water level data collected by sensors. In addition, Ulil et al. [80] developed pothole monitoring systems to detect potholes on the road caused by floods. Decision trees and machine support vector methods are the data analysis within the systems.

6. Comparison between the IoT Applied in Pre-Disaster and Post-Disaster Stages

We usually use sensors to collect environmental data in the pre-disaster stage. However, we should use sensors to collect health data and position data more in the postdisaster stage. We can transmit the data with a number of communication technologies and protocols in the pre-disaster stage. However, disasters may destroy communication infrastructures. Thus, it is important to maintain the efficiency of existing communication infrastructure and use additional mobile communication tools in the post-disaster stage. Machine learning techniques are common data analysis methods both in the pre-disaster stage and the post-disaster stage. With the development of image processing algorithms, we used cameras as sensors more. Fog-cloud computing is useful to improve the efficiency of data analysis both in the pre-disaster and post-disaster stages. The data applications aim to provide environmental information and send warnings in the pre-disaster stage. However, they focus more on route planning in the post-disaster stage.

In addition, five publications mention IoT systems used in both the pre-disaster and post-disaster stages [38,47,88,94,105]. Environmental data should be collected both in the pre-disaster and post-disaster stages since we need to use environmental data to detect disasters in the pre-disaster stage and ensure the safety of the environment in the post-disaster stage. For example, Rahman et al. [47] may use water level data to control the valves to prevent sewerage system overflow and mitigate floods. Since position data is also useful in the post-disaster stage, people can check environmental data and position data in Lwin et al. [88]'s application in the pre-disaster and post-disaster stages. Furthermore, Kim et al. [38] and Konomi et al. [105] provide evacuation route planning in the applications.

7. Conclusions and Future Work

Disasters have the potential to inflict harm upon human lives and result in significant economic and environmental repercussions, particularly in densely populated urban areas [6,7]. Given the substantial magnitude of losses incurred by urban disasters, it is imperative to enhance the efficacy of urban disaster management. Furthermore, the implementation of effective disaster management strategies is of utmost importance for urban areas to successfully attain the UNSDGs [12]. Sensors play a crucial role in the acquisition of data within IoT devices. They serve as a connection between the physical and digital realms, playing a vital function within the IoT framework. IoT has the potential to offer real-time monitoring, early warning systems, post-disaster response, and rescue support, thereby playing a significant role in the field of urban catastrophe management.

This study conducted a SLR to examine the utilization of sensors in the IoT for urban catastrophe management. The evaluation encompassed both the pre-disaster and post-disaster stages and analyzed a total of 72 publications. This study fills the research gap in sensors on IoT systems for urban disaster management. Nevertheless, this study possesses certain limitations that warrant further investigation in order to enhance its findings. The utilization of keywords has the potential to restrict the size of the sample. The formulation of keywords and establishment of inclusion criteria are based on the research questions, which may lead to the exclusion of some articles. Additionally, the selected publications may place a greater emphasis on the topics of earthquakes, landslides, and floods. There are a limited number of scholarly works that specifically address the topic of natural disasters, such as storms.

In addition, we explore potential avenues for future study in the field of disaster management. Specifically, we highlight the following areas: sensor heterogeneity, postdisaster emergency communication, integration of sensor technology with unmanned aerial vehicles, user participation in sensing, and the calculation of post-disaster rescue time.

Firstly, it can be observed that a diverse range of sensors are employed in the field of disaster management, encompassing both the pre-disaster and post-disaster stages. The presence of various manufacturers and the diverse applications of sensors in disaster scenarios contribute to the heterogeneity of these sensors, hence hindering the integration and sharing of information [107,108]. Some disasters may cause sequent disasters. For example, seismic activity or inundations can cause floods [109]. By performing an analysis of sensor data pertaining to various sorts of disasters, it becomes possible to anticipate the occurrence of subsequent disasters following an initial one. Moreover, future research endeavors could explore the integration of disparate sensors in order to develop a holistic application capable of facilitating the visualization of sensor data and the dissemination of alerts pertaining to various categories of disasters. To facilitate communication among heterogeneous sensors, future research endeavors may explore novel communication technologies and protocols, including the incorporation of integration brokerage applications. The JosNet system serves as a brokerage platform that facilitates interoperability and integration among many low-rate and low-power protocols, including Bluetooth LE, Zigbee, and Thread [110]. Furthermore, integration brokerage programs provide seamless communication with the remaining sensors and other heterogeneous devices throughout the post-disaster phase, hence enabling sustained communication in this critical time.

Secondly, in the aftermath of a disaster, it is possible to employ low-power communication technologies and protocols to sustain communication. For instance, one such technology is low-power satellite communication protocols. Random-access, very-lowpower, and wide-area networks (RA-vLPWANs), as low-power satellite communication protocols, provide uncoordinated multiple access in scenarios characterized by poor signalto-noise ratios and very low signal power [111]. Furthermore, the CEEA algorithm, as suggested by Ai-Turjman [102], serves the purpose of establishing a post-disaster sensor network by effectively connecting operational sensors with remaining energy. Notably, this algorithm bears a resemblance to the Bee algorithm [112]. Future research endeavors may be directed toward examining the potential applicability of the Bee algorithm in optimizing sensor networks during the post-disaster phase.

Thirdly, future studies also can combine sensors with unmanned aerial vehicles. We can install sensors on unmanned aerial vehicles. Future studies may investigate what types of sensors are suitable to install on unmanned aerial vehicles. If we use cameras on unmanned aerial vehicles, it is useful to investigate how to transmit the image data from unmanned aerial vehicles to the operation centers. Since we may control unmanned aerial vehicles remotely, how do we ensure the connects between controllers and unmanned aerial vehicles? Unmanned aerial vehicles can also serve as communication tools, especially in the post-disaster stage. It is interesting to optimize unmanned aerial vehicles' cruising trajectory to balance the communication coverage and cost. Future studies may also investigate how to use unmanned aerial vehicles to locate the sensors or trapped people.

Fourthly, some studies mentioned user participation in sensing in the pre-disaster stage. However, people may upload false information to the system, affecting the credibility of the system. Future studies may investigate how to ensure the authenticity of data with Blockchain. In addition, because of the wide application of cameras, sensor communication must be able to transmit a larger amount of image data.

Fifthly, future studies can use sensors to calculate the time from departure to the successful completion of the rescuers in the post-disaster stage, which can allocate rescue personnel more efficiently. For example, personal devices can record the rescuers' routes and calculate the time. The rescuers may record the completion of the rescue when they save the people successfully.

Author Contributions: Writing—original draft preparation, F.Z.; writing—review and editing, C.P. and H.T. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Blasi, S.; Ganzaroli, A.; De Noni, I. Smartening sustainable development in cities: Strengthening the theoretical linkage between smart cities and SDGs. *Sustain. Cities Soc.* 2022, *80*, 103793. [CrossRef]
- Ullah, Z.; Naeem, M.; Coronato, A.; Ribino, P.; De Pietro, G. Blockchain Applications in Sustainable Smart Cities. Soc. 2023, 97, 104697. [CrossRef]
- 3. Buyukozkan, G.; Ilicak, O.; Feyzioglu, O. A review of urban resilience literature. *Sustain. Cities Soc.* **2022**, *77*, 103579. [CrossRef]
- Xia, H.; Liu, Z.; Efremochkina, M.; Liu, X.; Lin, C. Study on city digital twin technologies for sustainable smart city design: A review and bibliometric analysis of geographic information system and building information modeling integration. *Sustain. Cities Soc.* 2022, *84*, 104009. [CrossRef]
- Ahmad, R.F.; Malik, A.S.; Qayyum, A.; Kamel, N. Disaster Monitoring in Urban and Remote Areas using Satellite Stereo Images: A Depth Estimation Approach. In Proceedings of the IEEE 11th International Colloquium on Signal Processing & Its Applications (CSPA), Kuala Lumpur, Malaysia, 6–8 March 2015; IEEE: Piscataway, NJ, USA, 2015; pp. 150–155.
- Tang, H.; Elalouf, A.; Levner, E.; Cheng, T.C.E. Efficient computation of evacuation routes on a three-dimensional geometric network. *Comput. Ind. Eng.* 2014, 76, 231–242. [CrossRef]
- Zhang, Q.X.; Hu, J.Y.; Song, X.P.; Li, Z.H.; Yang, K.H.; Sha, Y.Z. How does social learning facilitate urban disaster resilience? A systematic review. *Environ. Hazards-Hum. Policy Dimens.* 2020, 19, 107–129. [CrossRef]
- Cao, Y.; Xu, C.; Aziz, N.M.; Kamaruzzaman, S.N. BIM–GIS Integrated Utilization in Urban Disaster Management: The Contributions, Challenges, and Future Directions. *Remote Sens.* 2023, 15, 1331. [CrossRef]
- 9. Shah, S.A.; Seker, D.Z.; Rathore, M.M.; Hameed, S.; Ben Yahia, S.; Draheim, D. Towards Disaster Resilient Smart Cities: Can Internet of Things and Big Data Analytics Be the Game Changers? *IEEE Access* **2019**, *7*, 91885–91903. [CrossRef]
- Ali, K.; Nguyen, H.X.; Vien, Q.T.; Shah, P.; Raza, M.; Paranthaman, V.V.; Er-Rahmadi, B.; Awais, M.; ul Islam, S.; Rodrigues, J. Review and Implementation of Resilient Public Safety Networks: 5G, IoT, and Emerging Technologies. *IEEE Netw.* 2021, 35, 18–25. [CrossRef]
- Munawar, H.S.; Mojtahedi, M.; Hammad, A.W.A.; Kouzani, A.; Mahmud, M.A.P. Disruptive technologies as a solution for disaster risk management: A review. *Sci. Total Environ.* 2022, 806, 15. [CrossRef]
- Chen, M.; Lu, Y.J.; Peng, Y.; Chen, T.T.; Zhang, Y.Y. Key Elements of Attentions for Enhancing Urban Resilience: A Comparison of Singapore, Hong Kong and Hangzhou. *Buildings* 2022, 12, 340. [CrossRef]
- 13. Sinha, A.; Kumar, P.; Rana, N.P.; Islam, R.; Dwivedi, Y.K. Impact of internet of things (IoT) in disaster management: A task-technology fit perspective. *Ann. Oper. Res.* 2019, 283, 759–794. [CrossRef]
- 14. Esposito, M.; Palma, L.; Belli, A.; Sabbatini, L.; Pierleoni, P. Recent Advances in Internet of Things Solutions for Early Warning Systems: A Review. *Sensors* 2022, 22, 2124. [CrossRef] [PubMed]
- 15. Ochoa, S.F.; Santos, R. Human-centric wireless sensor networks to improve information availability during urban search and rescue activities. *Inf. Fusion* 2015, 22, 71–84. [CrossRef]
- Tudose, D.S.; Deaconu, I.; Musat, A. Geo-dynamic monitoring using wireless sensor networks. In Proceedings of the 15th RoEduNet Conference—Networking in Education and Research, Bucharest, Romania, 7–9 September 2016; IEEE: Piscataway, NJ, USA, 2016.
- 17. Sethi, P.; Sarangi, S.R. Internet of Things: Architectures, Protocols, and Applications. J. Electr. Comput. Eng. 2017, 2017, 9324035. [CrossRef]
- Amodu, O.A.; Nordin, R.; Jarray, C.; Bukar, U.A.; Mahmood, R.A.R.; Othman, M. A Survey on the Design Aspects and Opportunities in Age-Aware UAV-Aided Data Collection for Sensor Networks and Internet of Things Applications. *Drones* 2023, 7, 260. [CrossRef]
- de Camargo, E.T.; Spanhol, F.A.; Slongo, J.S.; da Silva, M.V.R.; Pazinato, J.; Lobo, A.V.D.; Coutinho, F.R.; Pfrimer, F.W.D.; Lindino, C.A.; Oyamada, M.S.; et al. Low-Cost Water Quality Sensors for IoT: A Systematic Review. *Sensors* 2023, 23, 4424. [CrossRef] [PubMed]
- Ahmed, T.; Creedon, L.; Gharbia, S.S. Low-Cost Sensors for Monitoring Coastal Climate Hazards: A Systematic Review and Meta-Analysis. *Sensors* 2023, 23, 1717. [CrossRef]
- Qadir, Z.; Ullah, F.; Munawar, H.S.; Al-Turjman, F. Addressing disasters in smart cities through UAVs path planning and 5G communications: A systematic review. *Comput. Commun.* 2021, 168, 114–135. [CrossRef]
- 22. Tei, A.I.; Doukha, Z.; Zafoune, Y. 3DMAT: Data dissemination for disaster management using available technology in a smart city. *Ann. Télécommun.* **2021**, *76*, 633–645. [CrossRef]
- 23. Snyder, H. Literature review as a research methodology: An overview and guidelines. J. Bus. Res. 2019, 104, 333–339. [CrossRef]

- 24. Xu, G.; Shi, Y.; Sun, X.; Shen, W. Internet of Things in Marine Environment Monitoring: A Review. *Sensors* 2019, *19*, 1711. [CrossRef] [PubMed]
- 25. Haaland, C.; van den Bosch, C.K. Challenges and strategies for urban green-space planning in cities undergoing densification: A review. *Urban For. Urban Green.* **2015**, *14*, 760–771. [CrossRef]
- Kim, S.W.; Brown, R.D. Urban heat island (UHI) intensity and magnitude estimations: A systematic literature review. *Sci. Total Environ.* 2021, 779, 146389. [CrossRef]
- 27. Meerow, S.; Newell, J.P.; Stults, M. Defining urban resilience: A review. Landsc. Urban Plan. 2016, 147, 38–49. [CrossRef]
- 28. Yigitcanlar, T.; Desouza, K.; Butler, L.; Roozkhosh, F. Contributions and Risks of Artificial Intelligence (AI) in Building Smarter Cities: Insights from a Systematic Review of the Literature. *Energies* **2020**, *13*, 1473. [CrossRef]
- 29. Zhang, Y.; Moyle, B.; Dupré, K.; Lohmann, G.; Desha, C.; MacKenzie, I. Tourism and natural disaster management: A systematic narrative review. *Tour. Rev. (Assoc. Int. d'Experts Sci. Tour.)* 2023, *ahead-of-print.* [CrossRef]
- Iqbal, U.; Perez, P.; Barthelemy, J. A process-driven and need-oriented framework for review of technological contributions to disaster management. *Heliyon* 2021, 7, e08405. [CrossRef]
- Cicek, D.; Kantarci, B. Use of Mobile Crowdsensing in Disaster Management: A Systematic Review, Challenges, and Open Issues. Sensors 2023, 23, 1699. [CrossRef]
- Kaur, M.; Kaur, P.D.; Sood, S.K. ICT in disaster management context: A descriptive and critical review. *Environ. Sci. Pollut. Res. Int.* 2022, 29, 86796–86814. [CrossRef]
- 33. Oktari, R.S.; Munadi, K.; Idroes, R.; Sofyan, H. Knowledge management practices in disaster management: Systematic review. *Int. J. Disaster Risk Reduct.* **2020**, *51*, 101881. [CrossRef]
- 34. Ritchie, B.W.; Jiang, Y. A review of research on tourism risk, crisis and disaster management: Launching the annals of tourism research curated collection on tourism risk, crisis and disaster management. *Ann. Tour. Res.* **2019**, *79*, 102812. [CrossRef]
- 35. Mishra, D.; Kumar, S.; Hassini, E. Current trends in disaster management simulation modelling research. *Ann. Oper. Res.* 2019, 283, 1387–1411. [CrossRef]
- 36. Goldschmidt, K.H.; Kumar, S. Humanitarian operations and crisis/disaster management: A retrospective review of the literature and framework for development. *Int. J. Disaster Risk Reduct.* **2016**, 20, 1–13. [CrossRef]
- 37. Furquim, G.; Pessin, G.; Faical, B.S.; Mendiondo, E.M.; Ueyama, J. Improving the accuracy of a flood forecasting model by means of machine learning and chaos theory. *Neural Comput. Appl.* **2016**, *27*, 1129–1141. [CrossRef]
- Kim, S.; Khan, I.; Choi, S.; Kwon, Y.W. Earthquake Alert Device Using a Low-Cost Accelerometer and its Services. *IEEE Access* 2021, 9, 121964–121974. [CrossRef]
- 39. Siringoringo, D.M.; Fujino, Y.; Suzuki, M. Long-term continuous seismic monitoring of multi-span highway bridge and evaluation of bearing condition by wireless sensor network. *Eng. Struct.* **2023**, *276*, 20. [CrossRef]
- 40. Acosta-Coll, M.; Ballester-Merelo, F.; Martinez-Peiro, M. Early warning system for detection of urban pluvial flooding hazard levels in an ungauged basin. *Nat. Hazards* **2018**, *92*, 1237–1265. [CrossRef]
- 41. Furquim, G.; Filho, G.P.R.; Jalali, R.; Pessin, G.; Pazzi, R.W.; Ueyama, J. How to Improve Fault Tolerance in Disaster Predictions: A Case Study about Flash Floods Using IoT, ML and Real Data. *Sensors* **2018**, *18*, 907. [CrossRef]
- Mendoza-Cano, O.; Aquino-Santos, R.; Lopez-de la Cruz, J.; Edwards, R.M.; Khouakhi, A.; Pattison, I.; Rangel-Licea, V.; Castellanos, E.; Martinez-Preciado, M.A.; Rincon-Avalos, P.; et al. Experiments of an IoT-based wireless sensor network for flood monitoring in Colima, Mexico. J. Hydroinform. 2021, 23, 385–401. [CrossRef]
- Siek, M.; Larry, L. Design and Implementation of Internet of Things and Cloud Technology in Flood Risk Mitigation. In Proceedings of the 3rd International Conference on Cybernetics and Intelligent System (ICORIS), Makassar, Indonesia, 25–26 October 2021; IEEE: Piscataway, NJ, USA, 2021; pp. 516–521.
- Malik, H.; Kandler, N.; Alam, M.M.; Annus, I.; Moullec, Y.; Kuusik, A. Evaluation of Low Power Wide Area Network Technologies for Smart Urban Drainage Systems. In Proceedings of the IEEE International Conference on Environmental Engineering (EE), Milan, Italy, 12–14 March 2018; IEEE: Piscataway, NJ, USA, 2018.
- 45. Furquim, G.; Neto, F.; Pessin, G.; Ueyama, J.; de Albuquerque, J.P.; Clara, M.; Mendiondo, E.M.; de Souza, V.C.B.; de Souza, P.; Dimitrova, D.; et al. Combining wireless sensor networks and machine learning for flash flood nowcasting. In Proceedings of the 28th IEEE International Conference on Advanced Information Networking and Applications Workshops (IEEE WAINA), Victoria, BC, Canada, 13–16 May 2014; pp. 67–72.
- Dai, W.J.; Tang, Y.N.; Zhang, Z.Y.; Cai, Z.M. Ensemble Learning Technology for Coastal Flood Forecasting in Internet-of-Things-Enabled Smart City. Int. J. Comput. Intell. Syst. 2021, 14, 166. [CrossRef]
- Rahman, M.M.; Abul Kashem, M.; Mohiuddin, M.; Hossain, M.A.; Moon, N.N. Future City of Bangladesh: IoT Based Autonomous Smart Sewerage and Hazard Condition Sharing System. In Proceedings of the 6th IEEE International Women in Engineering (WIE) Conference on Electrical and Computer Engineering (WIECON-ECE), Bhubaneswar, India, 26–27 December 2020; IEEE: Piscataway, NJ, USA, 2020; pp. 138–142.
- Garcia, V.M.; Granados, R.P.; Medina, M.E.; Ochoa, L.; Mondragon, O.A.; Cheu, R.L.; Villanueva-Rosales, N.; Rosillo, V.M.L. Management of real-time data for a smart flooding alert system. In Proceedings of the IEEE International Smart Cities Conference (ISC2), virtual, 28 September–1 October 2020; IEEE: Piscataway, NJ, USA, 2020.
- 49. Dhaya, R.; Ahanger, T.A.; Asha, G.R.; Ahmed, E.A.; Tripathi, V.; Kanthavel, R.; Atiglah, H.K. Cloud-Based IoE Enabled an Urban Flooding Surveillance System. *Comput. Intell. Neurosci.* 2022, 2022, 8470496. [CrossRef]

- Ragnoli, M.; Stornelli, V.; Del Tosto, D.; Barile, G.; Leoni, A.; Ferri, G. Flood monitoring: A LoRa based case-study in the city of L'Aquila. In Proceedings of the 17th Conference on Ph.D Research in Microelectronics and Electronics (PRIME), Villasimius, Italy, 12–15 June 2022; IEEE: Piscataway, NJ, USA, 2022; pp. 329–332.
- Biabani, M.; Fotouhi, H.; Yazdani, N. An Energy-Efficient Evolutionary Clustering Technique for Disaster Management in IoT Networks. Sensors 2020, 20, 2647. [CrossRef]
- Katsikogiannis, P.; Zervas, E.; Kaltsas, G. A wireless sensor network for building structural health monitoring and seismic detection. In Proceedings of the 3rd International Conference on Micro-Nanoelectronics, Nanotechnology and MEMs, Athens, Greece, 18–21 November 2007; pp. 3834–3838.
- Taale, A.; Ventura, C.E.; Marti, J. On the feasibility of IoT-based smart meters for earthquake early warning. *Earthq. Spectra* 2021, 37, 2066–2083. [CrossRef]
- Castelli, F.; Lentini, V.; Ferraro, A.; Grasso, S. Seismic Risk Evaluation for the Emergency Management. Ann. Geophys. 2018, 61, 220. [CrossRef]
- 55. Santos, A.S.; Corsi, A.C.; Almeida, R.Z.H.; Noda, M.K.; Goncales, I.; Ribeiro, R.N.; Machado, C.O.; Polkorny, M.; Otero, M.D.; Abreu, A.E.S.; et al. Feasibility study for detecting shallow landslides using IoT devices in smart cities. In Proceedings of the IEEE International Smart Cities Conference (ISC2), Manchester, UK, 7–10 September 2021.
- Ciampalini, A.; Farina, P.; Lombardi, L.; Nocentini, M.; Taurino, V.; Guidi, R.; della Pina, F.; Tavarini, D. Integration of Satellite InSAR with a Wireless Network of Geotechnical Sensors for Slope Monitoring in Urban Areas: The Pariana Landslide Case (Massa, Italy). *Remote Sens.* 2021, 13, 2534. [CrossRef]
- 57. Wang, H.H.; Tuo, X.G.; Zhang, G.Y.; Peng, F.L. Panzhihua airport landslide (3 October 2009) and an emergency monitoring and warning system based on the internet of things. *J. Mt. Sci.* **2013**, *10*, 873–884. [CrossRef]
- 58. Miao, F.; Yuan, Q. A WebGIS-Based Information System for Monitoring and Warning of Geological Disasters for Lanzhou City, China. *Adv. Meteorol.* **2013**, 2013, 769270. [CrossRef]
- Covenas, F.E.M.; Palomares, R.; Milla, M.A.; Verastegui, J.; Cornejo, J. Design and development of a low-cost wireless network using IoT technologies for a mudslides monitoring system. In Proceedings of the IEEE URUCON Conference (IEEE URUCON), Montevideo, Uruguay, 24–26 November 2021; IEEE: Piscataway, NJ, USA, 2021; pp. 172–176.
- Zhang, T. Advanced Wavelet Sampling Algorithm for IoT based environmental monitoring and management. *Comput. Commun.* 2020, 150, 547–555. [CrossRef]
- 61. Ferraz, R.; das Neves, C.R.G.; Silva, J. Internet of things with Web technologies solution for flood notification in Sao Paulo. In Proceedings of the 5th Edition of The Global IoT Summit (GIoTS), Dublin, Ireland, 20–22 June 2022; pp. 124–134.
- 62. Wang, Q.H.; Abdelrahman, W. High-Precision AI-Enabled Flood Prediction Integrating Local Sensor Data and 3rd Party Weather Forecast. *Sensors* 2023, 23, 3065. [CrossRef] [PubMed]
- Hu, C.F.; Cheng, X.J.; Xiao, X.; Chen, Z.Q.; Wang, Z.H.; Xu, J.; Zhao, D.Z. Integrated application of water informatization: A case study from Zengcheng Guangzhou China. In Proceedings of the 6th International Conference on Agro-Geoinformatics, Fairfax, VA, USA, 7–10 August 2017; IEEE: Piscataway, NJ, USA, 2017; pp. 98–102.
- 64. Furquim, G.; Pessin, G.; Gomes, P.H.; Mendiondo, E.M.; Ueyama, J. A distributed approach to flood prediction using a WSN and ML: A comparative study of ML techniques in a WSN deployed in Brazil. In Proceedings of the 16th International Conference on Intelligent Data Engineering and Automated Learning (IDEAL), Wroclaw, Poland, 14–16 October 2022; pp. 485–492.
- 65. Loftis, J.D.; Forrest, D.; Katragadda, S.; Spencer, K.; Organski, T.; Nguyen, C.; Rhee, S. StormSense: A New Integrated Network of IoT Water Level Sensors in the Smart Cities of Hampton Roads, VA. *Mar. Technol. Soc. J.* **2018**, *52*, 56–67. [CrossRef] [PubMed]
- Gabbar, H.A.; Chahid, A.; Isham, M.U.; Grover, S.; Singh, K.P.; Elgazzar, K.; Mousa, A.; Ouda, H. HAIS: Highways Automated-Inspection System. *Technologies* 2023, 11, 51. [CrossRef]
- Depetris, I.L.; Romani, I.D.; Gonzalez, I.J.; Bottero, C.; Rui, M.; Buratto, V.; Lorenzati, J.; Vaira, D.; Rui, E. Control by IoT in Drinking Water Pumping Station and Cisterns. In Proceedings of the 19th Workshop on Information Processing and Control (RPIC), San Juan, Argentina, 3–5 November 2021; IEEE: Piscataway, NJ, USA, 2021.
- 68. Tyagi, V.; Rawat, N.; Ram, M. Reliability modelling and sensitivity analysis of IoT based flood alerting system. *J. Qual. Maint. Eng.* **2021**, *27*, 292–307. [CrossRef]
- Furquim, G.; Mello, R.; Pessin, G.; Faical, B.S.; Mendiondo, E.M.; Ueyama, J. An accurate flood forecasting model using wireless sensor networks and chaos theory: A case study with real WSN deployment in Brazil. In Proceedings of the 15th International Conference on Engineering Applications of Neural Networks (EANN), Techical University, Sofia, Bulgaria, 5–7 September 2014; pp. 92–102.
- Mousa, M.; Oudat, E.; Claudel, C. A novel dual traffic/flash flood monitoring system using passive infrared/ultrasonic sensors. In Proceedings of the 12th IEEE International Conference on Mobile Ad Hoc and Sensor Systems (IEEE MASS), Dallas, TX, USA, 19–22 October 2015; IEEE: Piscataway, NJ, USA, 2015; pp. 388–397.
- Liu, Z.Q.; Huang, J.; Wang, Q.F.; Wang, Y.B.; Fu, J. Real-time Barrier Lakes Monitoring and Warning System Based on Wireless Sensor Network. In Proceedings of the 4th International Conference on Intelligent Control and Information Processing (ICICIP), Beijing, China, 9–11 June 2013; IEEE: Piscataway, NJ, USA, 2013; pp. 551–554.

- 72. Castro, U.; Avila, J.; Sustaita, C.V.; Hernandez, M.A.; Larios, V.M.; Villanueva-Rosales, N.; Mondragon, O.; Cheu, R.L.; Maciel, R. Towards Smart Mobility During Flooding Events in Urban Areas using Crowdsourced Information. In Proceedings of the 5th IEEE Annual International Smart Cities Conference (ISC2), Université Hassan II de Casablanca, Casablanca, Morocco, 14–17 October 2019; IEEE: Piscataway, NJ, USA, 2019; pp. 154–159.
- Junior, F.E.F.; Nonato, L.G.; Ranieri, C.M.; Ueyama, J. Memory-Based Pruning of Deep Neural Networks for IoT Devices Applied to Flood Detection. Sensors 2021, 21, 7506. [CrossRef]
- 74. Gabriel, P.E.; Butt, S.A.; Francisco, E.O.; Alejandro, C.P.; Maleh, Y. Performance analysis of 6LoWPAN protocol for a flood monitoring system. *EURASIP J. Wirel. Commun. Netw.* **2022**, 2022, 16. [CrossRef]
- Park, M.S.; Baek, K. Quality Management System for an IoT Meteorological Sensor Network-Application to Smart Seoul Data of Things (S-DoT). Sensors 2023, 23, 2384. [CrossRef]
- Alhamidi; Pakpahan, V.H.; Simanjuntak, J.E.S. Analysis of tsunami disaster resilience in Bandar Lampung Bay Coastal Zone. In Proceedings of the 1st ITB Centennial and 4th PlanoCosmo International Conference, Institut Teknologi Bandung, Bandung, Indonesia, 2–5 April 2018, IOP: Bristol, UK, 2018.
- 77. Aljohani, S.L.; Alenazi, M.J.F. MPResiSDN: Multipath Resilient Routing Scheme for SDN-Enabled Smart Cities Networks. *Appl. Sci.* **2021**, *11*, 1900. [CrossRef]
- Viegas, D.X. SWeFS: Sensor Web Fire Shield for Forest Fire Detection and Monitoring; University of Coimbra: Coimbra, Portugal, 2014; pp. 1537–1547. [CrossRef]
- Wang, E.K.; Wang, F.; Kumari, S.; Yeh, J.H.; Chen, C.M. Intelligent monitor for typhoon in IoT system of smart city. *J. Supercomput.* 2021, 77, 3024–3043. [CrossRef]
- Ulil, A.M.R.; Fiannurdin, S.S.; Tjahjono, A.; Basuki, D.K. The Vehicle as a Mobile Sensor Network base IoT and Big Data for Pothole Detection Caused by Flood Disaster. In Proceedings of the 12th International Interdisciplinary Studies Seminar on Environmental Conservation and Education for Sustainable Development (IISS), Malang, Indonesia, 14–15 November 2018; IOP: Bristol, UK, 2018.
- Kanak, A.; Arif, I.; Kumas, O.; Ergun, S. Extending BIM to urban semantic context for data-driven crisis preparedness. In Proceedings of the IEEE International Conference on Systems, Man, and Cybernetics (SMC), Toronto, ON, Canada, 11–14 October 2020; IEEE: Piscataway, NJ, USA, 2020; pp. 3813–3818.
- 82. Dhaya, R.; Kanthavel, R. IoT based urban flooding high definition surveillance using concurrent multipath wireless system. *Earth Sci. Inform.* **2022**, *15*, 1407–1416. [CrossRef]
- Gomes, J.L.; Jesus, G.; Rogeiro, J.; Oliveira, A.; da Costa, R.T.; Fortunato, A.B. An innovative web platform for flood risk management. In Proceedings of the 3rd International Conference on Innovative Network Systems and Applications (iNetSApp) Held in Conjunction with Federated Conference on Computer Science and Information Systems (FedCSIS), Lodz, Poland, 13–16 September 2015; pp. 217–231.
- Luo, J.; Xu, L.; Jamont, J.P.; Zeng, L.; Shi, Z. Flood decision support system on agent grid: Method and implementation. *Enterp. Inf. Syst.* 2007, 1, 49–68. [CrossRef]
- Aljohani, F.H.; Sen, A.A.A.; Ramazan, M.S.; Alzahrani, B.; Bahbouh, N.M. A Smart Framework for Managing Natural Disasters Based on the IoT and ML. *Appl. Sci.* 2023, 13, 3888. [CrossRef]
- Chen, C.; Jiang, J.G.; Zhou, Y.; Lv, N.; Liang, X.X.; Wan, S.H. An edge intelligence empowered flooding process prediction using Internet of things in smart city. J. Parallel Distrib. Comput. 2022, 165, 66–78. [CrossRef]
- Goyal, H.R.; Ghanshala, K.K.; Sharma, S. Post flood management system based on smart IoT devices using AI approach. In Proceedings of the International Conference on Technological Advancements in Materials Science and Manufacturing (ICTAMSM), Dehradun, India, 19–20 February 2021; pp. 10411–10417.
- Lwin, K.K.; Sekimoto, Y.; Takeuchi, W.; Zettsu, K. City geospatial dashboard: IoT and big data analytics for geospatial solutions provider in disaster management. In Proceedings of the 6th International Conference on Information and Communication Technologies for Disaster Management (ICT-DM), ESIEE Paris, Paris, France, 18–20 December 2019.
- Malik, H.; Zatar, W. A Real-Time and Low-Cost Flash Flood Monitoring System to Support Transportation Infrastructure. In Proceedings of the 44th Annual IEEE-Computer-Society International Conference on Computers, Software, and Applications (COMPSAC), Madrid, Spain, 13–17 July 2020; pp. 1111–1112.
- Sahil, H.; Sood, S.K. Fog-Cloud centric IoT-based cyber physical framework for panic oriented disaster evacuation in smart cities. *Earth Sci. Inform.* 2022, 15, 1449–1470. [CrossRef]
- Suri, N.; Zielinski, Z.; Tortonesi, M.; Fuchs, C.; Pradhan, M.; Wrona, K.; Furtak, J.; Vasilache, D.B.; Street, M.; Pellegrini, V.; et al. Exploiting Smart City IoT for Disaster Recovery Operations. In Proceedings of the 4th IEEE World Forum on Internet of Things (WF-IoT), Singapore, 5–8 February 2018; pp. 458–463.
- Johnsen, F.T.; Zielinski, Z.; Wrona, K.; Suri, N.; Fuchs, C.; Pradhan, M.; Furtak, J.; Vasilache, B.; Pellegrini, V.; Dyk, M.; et al. Application of IoT in military operations in a smart city. In Proceedings of the International Conference on Military Communications and Information Systems (ICMCIS), Warsaw, Poland, 22–23 May 2018.
- Korkalainen, M.; Mayra, A.P.; Kansala, K. An open communication and sensor platform for urban search and rescue operations. In Proceedings of the Conference on Unmanned/Unattended Sensors and Sensor Networks IX, Edinburgh, UK, 26–27 September 2012.
- 94. Boukerche, A.; Coutinho, R.W.L. Smart disaster detection and response system for smart cities. In Proceedings of the IEEE Symposium on Computers and Communications (IEEE ISCC), Natal, Brazil, 25–28 June 2018; pp. 1107–1112.

- 95. Anagnostopoulos, T.; Ntalianis, K.; Tsapatsoulis, N. IoT-enabled ambulances assisting citizens' wellbeing after earthquake disasters in smart cities. In Proceedings of the IEEE International Congres on Cybermatics/12th IEEE International Conference on Cyber, Physical and Social Computing (CPSCom)/15th IEEE International Conference on Green Computing and Communications (GreenCom)/12th IEEE International Conference on Internet of Things (iThings)/5th IEEE International Conference on Smart Data, Atlanta, GA, USA, 14–17 July 2019; IEEE: Piscataway, NJ, USA, 2019; pp. 922–928.
- Goyal, H.R.; Ghanshala, K.K.; Sharma, S. Flash flood risk management modeling in indian cities using IoT based reinforcement learning. In Proceedings of the International Conference on Technological Advancements in Materials Science and Manufacturing (ICTAMSM), Dehradun, India, 19–20 February 2021.
- 97. Antonacci, E.; Ceci, A.M.; Colarieti, A.; Gattulli, V.; Graziosi, F.; Lepidi, M.; Potenza, F. Dynamic testing and health monitoring via wireless sensor networks in the post-earthquake assessment of structural conditions at L'Aquila. In Proceedings of the 8th International Conference on Structural Dynamics (EURODYN), Leuven, Belgium, 4–6 July 2011; pp. 2440–2447.
- Ehsani, B.; Karimi, H.; Bakhshi, A.; Aghsami, A.; Rabbani, M. Designing humanitarian logistics network for managing epidemic outbreaks in disasters using Internet-of-Things. A case study: An earthquake in Salas-e-Babajani city. *Comput. Ind. Eng.* 2023, 175, 108821. [CrossRef] [PubMed]
- 99. Liu, Z.H.; Wang, C. Design of Traffic Emergency Response System Based on Internet of Things and Data Mining in Emergencies. *IEEE Access* 2019, 7, 113950–113962. [CrossRef]
- Campioni, L.; Lenzi, R.; Poltronieri, F.; Pradhan, M.; Tortonesi, M.; Stefanelli, C.; Suri, N. MARGOT: Dynamic IoT resource discovery for HADR environments. In Proceedings of the IEEE Military Communications Conference (MILCOM), Norfolk, VA, USA, 12–14 November 2019.
- Alvarez, F.; Almon, L.; Radtki, H.; Hollick, M. Bluemergency: Mediating post-disaster communication systems using the Internet of Things and Bluetooth mesh. In Proceedings of the 9th Annual IEEE Global Humanitarian Technology Conference (IEEE GHTC), Seattle, WA, USA, 17–20 October 2019; IEEE: Piscataway, NJ, USA, 2019; pp. 29–36.
- Ai-Turjman, F. Cognitive routing protocol for disaster-inspired Internet of Things. *Futur. Gener. Comp. Syst.* 2019, 92, 1103–1115. [CrossRef]
- 103. Tan, S.K.; Munro, A. Adaptive probabilistic epidemic protocol for wireless sensor networks in an urban environment. In Proceedings of the 16th International Conference on Computer Communications and Networks, Honolulu, HI, USA, 13–16 August 2007; IEEE: Piscataway, NJ, USA, 2007; pp. 1105–1110.
- 104. Kristalina, P.; Pratiarso, A.; Badriyah, T.; Putro, E.D. A wireless sensor networks localization using geometric triangulation scheme for object tracking in urban search and rescue application. In Proceedings of the 2nd International Conference on Science in Information Technology (ICSITech), Balikpapan, Indonesia, 26–27 October 2016; pp. 254–259.
- 105. Konomi, S.; Wakasa, K.; Ito, M.; Sezaki, K. User Participatory Sensing for Disaster Detection and Mitigation in Urban Environments. In Proceedings of the 4th International Conference on Distributed, Ambient and Pervasive Interactions (DAPI) Held as Part of 18th International Conference on Human-Computer Interaction (HCI International), Toronto, ON, Canada, 17–22 July 2016; pp. 459–469.
- 106. Li, F.; Wang, H.D.; Diao, L.J.; Yen, I.L.; Bastani, F. Toward semi-automated role mapping for IoT systems in smart cities. In Proceedings of the 5th IEEE Annual International Smart Cities Conference (ISC2), Casablanca, Morocco, 14–17 October 2019; IEEE: Piscataway, NJ, USA, 2019; pp. 205–211.
- Hong, J.-H.; Shi, Y.-T. Integration of Heterogeneous Sensor Systems for Disaster Responses in Smart Cities: Flooding as an Example. *ISPRS Int. Geo-Inf.* 2023, 12, 279. [CrossRef]
- 108. Saad, M.H.M.; Hamdan, N.M.; Sarker, M.R. State of the Art of Urban Smart Vertical Farming Automation System: Advanced Topologies, Issues and Recommendations. *Electronics* **2021**, *10*, 1422. [CrossRef]
- Lozano, J.M.; Tien, I. Data collection tools for post-disaster damage assessment of building and lifeline infrastructure systems. *Int. J. Disaster Risk Reduct.* 2023, 94, 103819. [CrossRef]
- 110. Balota, J.E.; Kor, A.-L. Brokerage System for Integration of LrWPAN Technologies. Sensors 2022, 22, 1733. [CrossRef] [PubMed]
- Krondorf, M.; Bittner, S.; Plettemeier, D.; Knopp, A.; Wikelski, M. ICARUS—Very Low Power Satellite-Based IoT. Sensors 2022, 22, 6329. [CrossRef] [PubMed]
- 112. Ang, M.C.; Ng, K.W. Minimising printed circuit board assembly time using the bees algorithm with TRIZ-inspired operators. In *Intelligent Production and Manufacturing Optimisation—The Bees Algorithm Approach*; Pham, D.T., Hartono, N., Eds.; Springer International Publishing: Cham, Switzerland, 2023; pp. 25–41. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.